

## Format for ANSWERING REVIEWERS



February 6, 2015

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 16007-Revised manuscript WJG.doc).

**Title: Anterior rectopexy for full-thickness rectal prolapse: technical and functional results**

**Author:** Jean-Luc Faucheron, Bertrand Trilling, Edouard Girard, Pierre-Yves Sage, Sandrine Barbois, Fabian Reche

**Name of Journal:** *World Journal of Gastroenterology*

**ESPS Manuscript NO:** 16007

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewers. The manuscript has been corrected following all reviewers comments (see first comments for the first review which was #11548)

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

Jean-Luc FAUCHERON, MD, PhD  
Colorectal Unit, Head of the Department of Surgery  
TIMC IMAG Laboratory  
Joseph Fourier University  
Grenoble, CS 10217  
FRANCE  
Fax: +33-4-76-76-87-80  
E-mail: JLFaucheron@chu-grenoble.fr